Author's response to reviews

Title: Uterotonic use immediately following birth: Using a novel methodology to estimate population coverage in four countries

Authors:

Jim Ricca (Jim.ricca@jhpiego.org)
Vikas Dwivedi (vdwivedi@jsi.com)
John Varallo (John.varallo@jhpiego.org)
Gajendra Singh (Gajendra.Singh@jhpiego.org)
Saranjeen P Pallipamula (Saranjeen.Pallipamula@jhpiego.org)
Nazir Amade (naibrahimo@yahoo.com.br)
Maria de Luz Vaz (Maria.vaz@jhpiego.org)
Dustan Bishanga (dustan.bishanga@jhpiego.org)
Marya Plotkin (Marya.plotkin@jhpiego.org)
Bashra Al-Makaleh (Bushra.Al-Makaleh@jhpiego.org)
Stephanie Suhowatsky (Stephanie.suhowatsky@jhpiego.org)
Jeffrey M Smith (Jeffrey.Smith@jhpiego.org)

Version: 3 Date: 24 November 2014

Author's response to reviews: see over
<table>
<thead>
<tr>
<th>REVIEWER'S REPORT</th>
<th>RESPONSE</th>
</tr>
</thead>
</table>
| **Title:** Uterotonic use immediately following birth: Using a novel methodology to estimate population coverage in four countries  
**Version:** 2  
**Date:** 27 October 2014  
**Reviewer:** Mercedes Colomar |

**Reviewer's report:**

<table>
<thead>
<tr>
<th>Minor essential revisions:</th>
</tr>
</thead>
<tbody>
<tr>
<td>1. please provide the sources where the methods for UUIFB estimation were taken</td>
</tr>
<tr>
<td>A column has been added to Table 1 (note, in the previous version Table 1 was referred to as Figure 1)</td>
</tr>
<tr>
<td>2. Line 347, I think the first sentence belongs to the previous paragraph.</td>
</tr>
<tr>
<td>This sentence was moved to a more appropriate place in the preceding paragraph where there was other information about how the observational assessment findings fed into adjustments.</td>
</tr>
</tbody>
</table>
| 3. Even though you have described and stated the number of experts participating in each panel, it would be important to provide the number of participants by position for each country.  
IE: Figure 2. Number and description of experts participating in each country panel  
An example for the figure I recommend will be uploaded |
| As suggested a table (Table 2) has been added. |

**Reviewer's report**

| **Title:** Uterotonic use immediately following birth: Using a novel methodology to estimate population coverage in four countries  
**Version:** 2  
**Date:** 28 October 2014  
**Reviewer:** Jane Hirst |

**Reviewer's report:**

This revised manuscript has addressed my initial queries and I have no further comments.